Author’s response to reviews

Title: The association between secondhand smoke and the risk of developing acute coronary syndromes, among non-smokers, under the presence of several cardiovascular risk factors: the CARDIO2000 case-control study

Authors:

Demosthenes Panagiotakos (dbpanag@math.uoa.gr)
Christina Chrysohoou (chrysohoou@usa.net)
Christos Pitsavos (chrysohoou@usa.net)
Ioanna Papaioannou (dbpanag@math.uoa.gr)
John Skoumas (chrysohoou@usa.net)
Christodoulos Stefanadis (chrysohoou@usa.net)
Pavlos Toutouzas (chrysohoou@usa.net)

Version: 9 Date: 23 May 2002

Dear Editor

According to your last e-mail I would like to inform you that:
1. We have removed the highlighted words from the text
2. We have removed the words indicating where the tables and figures are to be inserted
3. There is no any unnecessary space surrounding the image

There are no other comments from the reviewers.

Thank you very much for the attention given to our work.

Sincerely

D.B. Panagiotakos